

Citation: Hoehn KB, Pybus OG, Kleinstein SH (2022) Phylogenetic analysis of migration, differentiation, and class switching in B cells. PLoS Comput Biol 18(4): e1009885. https://doi.org/ 10.1371/journal.pcbi.1009885

Editor: Andrew J. Yates, Columbia University Medical Center: Columbia University Irving Medical Center, UNITED STATES

Received: August 16, 2021

Accepted: January 31, 2022

Published: April 25, 2022

Copyright: © 2022 Hoehn et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The methods developed in this paper are available in the R package dowser, available at https://dowser. readthedocs.io as part of the Immcantation suite (http://immcantation.org). Scripts for performing analyses in this manuscript are available at: https:// bitbucket.org/kleinstein/projects. Data from simulations are available at: https://doi.org/10. 5281/zenodo.5171523.

Funding: K.B.H was supported through a PhRMA Foundation post-doctoral fellowship in informatics

RESEARCH ARTICLE

Phylogenetic analysis of migration, differentiation, and class switching in B cells

Kenneth B. Hoehn¹, Oliver G. Pybus^{2,3}, Steven H. Kleinstein^{1,4,5}*

 Department of Pathology, Yale School of Medicine, New Haven, Connecticut, United States of America,
Department of Zoology, University of Oxford, Oxford, United Kingdom, 3 Department of Pathobiology and Population Sciences, Royal Veterinary College London, London, United Kingdom, 4 Program in Computational Biology and Bioinformatics, Yale University, New Haven, Connecticut, United States of America, 5 Department of Immunobiology, Yale School of Medicine, New Haven, Connecticut, United States of America

* steven.kleinstein@yale.edu

Abstract

B cells undergo rapid mutation and selection for antibody binding affinity when producing antibodies capable of neutralizing pathogens. This evolutionary process can be intermixed with migration between tissues, differentiation between cellular subsets, and switching between functional isotypes. B cell receptor (BCR) sequence data has the potential to elucidate important information about these processes. However, there is currently no robust, generalizable framework for making such inferences from BCR sequence data. To address this, we develop three parsimony-based summary statistics to characterize migration, differentiation, and isotype switching along B cell phylogenetic trees. We use simulations to demonstrate the effectiveness of this approach. We then use this framework to infer patterns of cellular differentiation and isotype switching from high throughput BCR sequence datasets obtained from patients in a study of HIV infection and a study of food allergy. These methods are implemented in the R package *dowser*, available at https://dowser.readthedocs.io.

Author summary

B cells produce high affinity antibodies through an evolutionary process of mutation and selection during adaptive immune responses. Migration between tissues, differentiation to cellular subtypes, and switching between different antibody isotypes can be important factors in shaping the role B cells play in response to infection, autoimmune disease, and allergies. B cell receptor (BCR) sequence data has the potential to elucidate important information about these processes. However, there is currently no robust, generalizable framework for making such inferences from BCR sequence data. Here, we develop three parsimony-based summary statistics to characterize migration, differentiation, and isotype switching along B cell phylogenetic trees. We confirm the effectiveness of our approach using simulations and further apply our method to data from patients with HIV and allergy. Our methods are released in the R package *dowser*: https://dowser.readthedocs.io.

(phrmafoundation.org). This work was funded in part by the European Research Council under the European Union's Seventh Framework Programme (FP7/2007-2013)/European Research Council grant agreement number 614725-PATHPHYLODYN (erc.europa.eu). This work was funded in part by National Institutes of Health, National Institute of Allergy and Infectious Diseases grant R01 Al104739 (niaid.nih.gov). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: S.H.K. receives consulting fees from Northrop Grumman and Peraton. K.B.H receives consulting fees from Prellis Biologics.

Introduction

The adaptive immune system in humans depends on B cells to produce antibodies capable of neutralizing a wide array of pathogens. Antibody structures are initially expressed as B cell receptors (BCRs) on the surfaces of B cells. BCRs are generated through random V(D)J recombination and then subjected to repeated rounds of somatic hypermutation (SHM), cell proliferation, and selection for antigen binding [1]. This evolutionary process, called affinity maturation, creates many lineages of B cells that each descend from a single naive progenitor cell. Cells within a clonal lineage differ predominately by point mutations. The genetic variation within these clonal lineages has long been investigated using phylogenetic methods [2]. High throughput sequencing of BCR sequences has shown promise in elucidating information about the adaptive immune response in humans, such as the sequence of mutations that occur during antibody co-evolution with HIV [3], and the process of mutation and selection during affinity maturation generally [4]. Other important biological processes may occur as BCR sequences evolve, such as B-cell migration between tissues [5], differentiation into cellular subsets [6], and antibody isotype class switching [7]. If these processes co-occur with SHM, then in principle they can be investigated and inferred using phylogenetic techniques.

Understanding the ways in which B cells migrate between tissues, differentiate into cell types, and switch isotypes can provide important information about the mechanism and treatment of B cell mediated diseases. For instance, previous work showed that B cells found in the brain lesions of multiple sclerosis patients arise from the cervical lymph nodes, which can be more easily targeted for treatment [5]. Other work in the autoimmune disease myasthenia gravis indicates that pathogenic B cells migrate from the thymus to the peripheral blood, suggesting they may be targeted for treatment with early thymectomy [8]. In the study of food allergies, B cell repertoire analysis has been used to show that allergy-causing IgE cells likely originate from distinct memory B cell subsets [9], and from sequential switching through IgG or IgA rather than from IgM [10]. Further work has shown that IgE producing B cells likely undergo isotype switching locally in gut associated tissue [11]. All of these processes have the potential to be studied using a combination of BCR repertoire analysis and molecular phylogenetic methods, so long as SHM occurs between transition events. This SHM could occur either through germinal center reactions, or through extrafollicular activation and SHM [12]. Possible scenarios in which migration among tissues could be studied in this way are shown in S1 Fig.

Migration and cellular differentiation in B cells can be viewed as analogous to geographic spread of rapidly evolving viruses, the study of which–viral phylogeography–has advanced both in theory and application in the past decade (e.g. [13]). For example, phylogeographic methods have been used to determine the origin of the HIV pandemic [14], factors influencing the recent Ebola epidemic [15,16], and the epidemic spread of Zika virus [17,18]. Modern phylogeographic analyses typically model phylogenetic sequence evolution [19], and changes in geographic location within a unified framework [20,21]. Successfully developing a phylogeographic framework for B cell lineages would enable the testing of new hypotheses regarding the nature of evolution during affinity maturation.

There are serious challenges that must be addressed before using modern phylogeographic methods on B cell repertoire datasets. Such techniques typically rely on molecular clock trees, whose branch lengths represent elapsed time between nodes [20]. Accurately modeling sequence change through time requires either data sampled at multiple time points, or prior information about the expected rate of sequence evolution. These are frequently not available for B cell lineages. Data samples, particularly biopsies, are often only taken at a single time point [5,22], and the variation of B cell mutation rate over time is largely unknown and likely

dependent on cell subset. Even using a Markov model to describe state changes along B cell molecular phylogenies is not straightforward: B cell lineage trees frequently contain identical sequences with different states. This results in state changes across zero-length branches, which do not fit within a Markov model framework. Further, modern phylogeographic techniques often rely on Markov chain Monte Carlo sampling, which makes them computationally intensive and impractical to apply to thousands of sequences. B cell bulk repertoires often contain millions of sequences, and individual lineages sometimes contain thousands of unique sequences.

We propose that hypotheses about B cell migration and differentiation may be usefully investigated using heuristic summary statistics that characterize the distribution of trait values along phylogenetic trees. Indeed, such heuristic approaches which do not depend on branch length have historically been used to test hypotheses about migration between populations [23–25]. While tree based summary statistics have been previously used to assess B cell migration [5], differentiation [6], and isotype switching [7], these approaches have not been tested through simulations and their general accuracy is unclear. To address this methodological gap, we develop a set of maximum parsimony-based statistics that summarize the relative distribution of B cell states along the tips of lineage trees within repertoires and introduce a framework for assessing the significance of their difference from randomized trees. We demonstrate through simulations that these tests relate intuitively to different regimes of migration and differentiation. To demonstrate its utility, we use this framework to test hypotheses regarding differentiation of cell types in HIV infection, and sequential class switching to IgE and IgG4. We introduce a statistically principled and scalable means of analyzing the evolution of discrete traits in B cell repertoires. We release these methods in the R package *dowser*.

Methods

Predicting states of internal tree nodes

The goal of the discrete trait analysis framework presented here is to characterize the distribution of predicted trait values along the tips of B cell lineage trees. Given an alignment of sequences inferred to descend from the same naive ancestor (i.e. the same clonal lineage), lineage tree topologies and branch lengths were estimated using maximum parsimony using *dnapars* v3.967 [26]. While other methods are available for inferring B cell lineage trees, maximum parsimony has been shown to be effective in some cases [27] and is faster than more complex maximum likelihood models designed for B cell lineages [4,28]. Importantly, the statistics presented here are not limited to tree topologies inferred through maximum parsimony; maximum parsimony is used here primarily for computational expediency.

Maximum parsimony is also used to infer the discrete character states (e.g. cell subtype, isotype, tissue) of internal nodes, given a tree in which each tip is associated with a given character state. Nodes with different states from their immediate ancestors are counted as state changes. More specifically, internal node states were reconstructed using the Sankoff dynamic programming maximum parsimony algorithm [29], which, given a weight matrix for each type of state change, determines the minimum number of state changes that must be made along the tree given the states at the tips. The backtrace step of this algorithm can be used to determine a set of most parsimonious internal node states. Often there are multiple such maximum parsimony sets. To represent state changes across ambiguous internal node sets, trajectories with equal parsimony were randomly chosen in the backtrace step of the Sankoff algorithm, beginning at the root of the tree and moving towards the tips. This process was performed 100 times for each tree, and the mean of each type of state change was reported. Strictly bifurcating B cell lineage trees frequently have clusters of nodes separated by zerolength branches (soft polytomies), which represent a high degree of uncertainty in tree topology. This uncertainty in the order of bifurcating nodes can result in a potentially large number of uninformative state changes along the polytomy. Multiple steps were taken to minimize the effects of random polytomy resolution (S1 Appendix). Briefly, nodes within each polytomy were first re-ordered to minimize the number of state changes along the tree. To represent the uncertainty in the order of state changes, nodes within each polytomy were grouped together into separate subtrees according to their predicted state. These state-specific subtrees were then joined together in a balanced manner, ensuring that state changes could occur in any direction among the states contained within the polytomy (S1 Appendix).

Testing trait-phylogeny association

Analysis begins with a B-cell lineage tree topology with discrete character states (trait values) associated with each tip, and internal node states reconstructed through maximum parsimony. The goal of our discrete trait analysis framework is to determine how the distribution of discrete character states along the internal nodes of a tree differs from its expectation if traits are randomly distributed among the tips. The statistics introduced herein are shown graphically in Fig 1. More formally, if there are *m* possible discrete character states, and o_{ij} is the number of



Fig 1. Hypothetical phylogenetic trees used to illustrate tree/trait association statistics. Trait values at internal nodes of the tree are inferred using maximum parsimony given the trait values at the tips, which are shown using different colors and shapes. Two example state changes are highlighted. (**left**) *No association between tip-trait values and tree*: Distribution of traits across this tree is indistinguishable from randomly distributed traits by any statistic used. (**middle**) *Tip-trait values clustered in tree*: Cells with the same trait value are more closely related to each other in the tree, which will yield significantly fewer switches than in the same tree with permuted tips, and therefore a significantly low *PS* statistic. (**right**) *Asymmetric ancestor/descendant relationships among trait values*: All switches in this tree are from *A* to *B* (*SP* = 1), significantly higher than expected from the same tree with permuted tips. This indicates an asymmetric relationship between these states along the tree.

https://doi.org/10.1371/journal.pcbi.1009885.g001

state changes from type *i* to type *j*, the three statistics investigated are defined as:

$$PS = \sum_{i}^{m} \sum_{j,j \neq i}^{m} o_{ij} \tag{1}$$

$$SC_{ij} = o_{ij}$$
 (2)

$$SP_{ij} = \frac{o_{ij}}{\sum_{i}^{m} \sum_{j,j \neq i}^{m} o_{ij}}$$
(3)

The *PS* (parsimony score) statistic is the total number of state changes along a tree. The *SC* (switch count) statistic from *i* to *j* is the number of state changes from state *i* to *j*. The *SP* (switch proportion) statistic from *i* to *j* is the proportion of state changes from state *i* to *j*.

We calculate the significance of these three statistics using a permutation test. This is done by randomizing traits at the tips of the lineage tree, re-calculating each statistic on the permuted tree, and repeating for a specified number of replicates. For each replicate we calculate the difference between the statistic calculated on the observed tree and the same statistic calculated on the permuted tree. We refer to this difference as δ . If mean $\delta > 0$ (hereafter mean δ is indicated by δ), this indicates the statistic is on average higher in observed trees than in permuted trees. For a one-tailed test, we calculate the *p* value that $\delta > 0$ as the proportion of replicates in which $\delta \leq 0$. Similarly, we calculate the *p* value that $\delta < 0$ as the proportion of replicates in which $\delta \geq 0$. For a two-tailed test, we calculate the *p* value that $\delta > 0$ as the proportion of replicates in which $\delta < 0$, plus half of the replicates in which $\delta = 0$. We refer to the calculation of these *p* values for the statistics in **Eqs 1–3** as the *PS* test, *SC* test, and *SP* test, respectively. The process of performing the *SP* test is shown graphically in **S2 Fig**. The overall process is the same for all three tests–they only differ in the statistic calculated.

These three statistical tests capture different aspects of how the distribution of characters observed along a lineage tree differs from random association between tree topology and trait values. The *PS* test determines the extent to which trait values are clustered together within the tree. A significantly low *PS* statistic (i.e. $\delta < 0$, p < 0.05) indicates identical trait values are more closely clustered together within the tree than expected from random association between tree topology and trait values. By contrast, a significantly high *PS* statistic indicates identical trait values are less clustered together than expected by chance. Variations of this test were previously developed in [23] and applied in [30] to study spread of influenza.

While the *PS* test only determines a general association between trait values and tree topology, the *SC* and *SP* tests are both aimed at determining whether a particular trait value is more ancestral to another in the tree. A significantly high *SC* statistic (**Eq 2**) from state *i* to state *j* indicates a greater number of switches from state *i* to state *j* than expected from random association between tree topology and trait values. The *SC* test was used by [24] in the context of virus phylogeography, and by [6] to decompose the phylogenetic relationships among B cell subtypes within HIV infection. The *SC* test, however, is expected to have a high false negative rate because trees with randomized tip states often have more state changes events in general. Because of this, *SC* test δ values tend to be negative even when there is no polarity in the ancestor/descendant relationship among type *i* and *j* (**Fig 1**).

To normalize for differences in the total number of state changes between observed and permuted trees, we introduce the *SP* test. A significantly high *SP* statistic (Eq 3) from state *i* to state *j* indicates a greater proportion of switches from state *i* to state *j* than expected from a random distribution of trait values at the tips. In contrast to the *SC* test, a significant association between the tree and trait may exist, but only associations in which one trait is more often

ancestral to the other than expected will give rise to a significantly high *SP* statistic (Fig 1C). Further, the denominator of the *SP* statistic can be altered to test other hypotheses. For instance, to test whether a greater proportion of state changes to state *j* come immediately from state *i* than expected by chance, one can restrict the analysis to consider only state changes towards *j*.

Accounting for uncertainty in tree topology

To account for uncertainty in tree topology, we bootstrap the multiple sequence alignment of each clone [31]. This is performed by randomly sampling multiple sequence alignment columns with replacement. Lineage tree topology and branch length estimation then proceeds as before. Test statistics are calculated for each bootstrap replicate tree, and then for a single permutation of the traits at that tree's tips. Calculations for δ values proceed as before, but with each o_{ij} value indexed for each replicate. This procedure is very similar to that proposed in [32] and can in principle be extended to other sets of tree topologies, such the posterior distribution of tree topologies generated by MCMC sampling under Bayesian phylogenetic inference.

From trees to repertoires

B cell repertoire datasets often consist of hundreds or thousands of B cell lineage trees. Often hypotheses do not concern individual lineages, but instead the behavior of the collection of B cell lineages as a group. To characterize multiple B cell lineages, the observed and permuted summary statistics are summed across all lineages of a repertoire for each bootstrap replicate. Additionally, traits may be permuted among trees, which may increase statistical power and detect nonrandom association among trait types within trees.

Simulations

We tested the performance of the three proposed statistics using simulations based on B cell lineage trees estimated from an empirical dataset. Sequences were obtained from peripheral blood samples taken from one subject at ten time points, from eight days before to 28 days after influenza vaccination [33] (subject 420IV). Sequence preprocessing and clonal clustering are described in [4]. Sequences were down-sampled by 50%, and only clones with >10 unique sequences were retained. A total of 399 clones containing 11 to 370 (mean = 26.2) unique sequences remained. Tree topologies and branch lengths were estimated for each clone using dnapars v3.967 [26] via the R package alakazam v0.3.0 [34] and dowser v0.0.1. We simulated state changing down each tree using a Markov model parametrized by initial frequencies π for each state, relative rate parameters r_{ii} for each pair of possible states i and j, and r, the average rate of state changes per mutation per site. The mean value of this rate matrix was calculated as the sum of the diagonal elements weighted by their initial state frequencies. All values of the matrix were then divided by this mean and multiplied by r. This calibration was performed so that r^*l state change events are expected to occur across a branch of length l mutations/site under the initial state frequencies. Because the frequency of each state can change along the tree (i.e. the Markov chain is not necessarily at equilibrium), r may not necessarily equal the average rate of state change events further down the tree. For each tree, the state at the germline node was randomly drawn based on each state's π value. For each node after the germline, the rate matrix is multiplied by the node's ancestral branch length and exponentiated to give the probability of each state at that node, given the state at its immediate ancestor node. The state was then randomly chosen based on these probabilities. This process begins at the germline node and continues down the tree until each tip node has a state. Because each tip corresponds to a sequence, this forms a dataset of sequences paired with simulated discrete

characters. Internal node states are not included in the final simulated dataset used for analysis. Simulations were performed with two state models (*A* and *B*) that explored a large parameter space ($\pi_a = 0.5, 1; r_{ab} = 1, 10; r = 10, 25, 50, 100, 1000$), and four state models (*A*, *B*, *C*, *D*) that explored more complex patterns of state change at low overall rate (r = 10). Twenty simulation repetitions were performed for each parameter combination. Statistical tests were performed as described in **Methods**; however, to improve computational efficiency simulation analyses did not use bootstrapped multiple sequence alignments, and instead performed 100 permutations on a fixed maximum parsimony tree for each clone. Only clones with more than one state type were analyzed.

To benchmark the performance of the *SP* test under more extreme conditions, we performed similar simulations along large phylogenies with asymmetrical topologies. In these simulations, perfect ladder topologies were used, with each internal node bifurcating to at least one tip and at most one internal node. All branch lengths were set to a length of 0.001 mutations/site. Simulations using these tree topologies used two states and otherwise continued as detailed above.

Controlling the false positive rate using down-sampling

Initial analyses using simulations along large ladder phylogenies demonstrated that the SP test had a high false positive rate when the overall rate of switching was extremely low (S6 Fig). These problematic cases are identifiable as they result in trees with a high ratio of tips to predicted state changes (S7 Fig). We adjust for these cases by using a simple down-sampling algorithm. The goal of the down-sampling algorithm is to randomly draw a subset of the original data whose resulting phylogenetic tree cannot have a tip-to-state change ratio greater than a pre-specified value k. A tree containing m state types at its tips must have at least m -1 state changes along its edges. If a tree contains m state types and $k^*(m-1)$ tips, then the ratio of tips to state changes (which must be > m-1) along its branches can be no larger than k. For the desired tip-to-state change ratio of k, the algorithm proceeds by randomly sampling one tip of each state type (m) from a tree, and then randomly sampling $k^*(m-1)-m$ remaining sequences without replacement. This produces a tree with *m* states, $k^*(m-1)$ tips, and a maximum tip-tostate change ratio of k. Tree building and summary statistic calculation then proceed as above. This down-sampling is repeated each repetition. For computational expediency, this process can be repeated on tree topologies that have been already estimated. Tips are simply sampled from the tree with the remaining topology and branch lengths intact. This down-sampling algorithm was implemented in *dowser* v0.1.0, and all analyses presented that use this algorithm use dowser v0.1.0, R v4.0.3 [35], and ape v5.5–3 (www.github.com/emmanuelparadis/ape, commit 4fd8482). All other analyses used dowser v0.0.1, R v3.6.1, and ape v0.5.3.

Empirical datasets

We demonstrated the utility of the proposed discrete trait framework by analyzing two empirical datasets. The first was aimed at understanding B cell differentiation during HIV infection, and consisted of BCR mRNA sequences taken from sorted populations of unswitched memory B cell (MBC), CD19^{hi} MBC, CD19^{lo} MBC, and germinal center B cells (GCBC) from three HIV viremic subjects (subject 1–3; [6]). Each dataset was subsampled to a maximum of 50,000 total sequences, and only clones with more than 10 sequences were retained. Unique sequences associated with more than one cell type were kept distinct. This resulted in 138, 203, and 166 multi-cell type clones with a mean of 49.9, 39.4, and 30.3 unique sequences per clone, for subjects 1–3 respectively. State changes across all lineages for each subject were calculated over 100 bootstrap replicates. To control the false positive rate of the *SP* test, all trees were down-sampled to a maximum tip-to-state change ratio of 20.

The second dataset was aimed at understanding isotype switching patterns in human children, and consists of BCR mRNA sequences obtained from peripheral blood samples taken from a human subject each year from age 1 to 3 years old [36]. Preprocessing, including grouping of sequences into clonal clusters, is detailed in <u>S2 Appendix</u>. Only clones with at least 4 unique sequences and more than one isotype were retained. Identical sequences associated with more than one isotype were kept distinct so each sequence was associated with one isotype. This resulted in 792 multi-istoype clones with a mean of 9.1 unique sequences each. State changes across all lineages were calculated over 100 bootstrap replicates. To control the false positive rate of the *SP* test, all trees were down-sampled to a maximum tip-to-state change ratio of 20.

Results

We outline three parsimony-based summary statistics to characterize the distribution of trait values along B cell lineage trees (Fig 1). The significance of these statistics is tested by comparing observed values within the set of trees that comprise a repertoire to those obtained from permuting trait values at the trees' tips. The mean difference between each statistic calculated from the observed trees and permuted trees is referred to as δ . The process of calculating δ and p values is shown graphically for the SP test in S2 Fig. The first statistic, the parsimony score (PS), is the total number of trait value state changes that occurred along a lineage tree. A PS test with δ < 0 and p < 0.05 (*i.e.* a significantly low PS statistic) indicates the trait values cluster together in the observed trees more often than expected by chance (Fig 1). We propose two other statistics aimed at determining whether one state is more frequently the immediate ancestor to another state than expected by chance. The switch count (SC) from state *i* to *j* is the number of state changes that occurred from *i* to *j* [24], while the switch proportion (SP) from state *i* to *j* is the proportion of state changes that occurred from *i* to *j*. An SC or SP test from *i* to *j* with $\delta > 0$ and p < 0.05 indicates trait value *i* was more frequently the immediate ancestor to state *j* than expected by chance. We expect the SP test to be more sensitive to this relationship than the SC test because it accounts for the higher number of state changes expected in randomized trees (Fig 1). Similarly, an SP test from *i* to *j* with $\delta < 0$ and p < 0.05 indicates trait value *i* was less ancestral to state *i* than expected. All three of these tests may be used to characterize individual lineages or entire B cell repertoires; in this paper we will focus exclusively on repertoires.

Differentiating state change patterns with two states

We used simulations to assess the performance of our proposed tests. We model B cell migration and differentiation using a Markov model with two states, A and B, and empiricallyderived lineage tree topologies (Methods). Briefly, the pattern of state changes along a tree was determined by the probability that the state at the root was A ($\pi_a = 0.5$, 1; $\pi_b = 1 - \pi_a$), the average rate of state change (r = 10, 25, 50, 100, 1000 changes/mutation/site), and the relative rate of change from A to B ($r_{ab} = 1, 10$; $r_{ba} = 1/r_{ab}$). These parameters represent a range of slow, fast, biased, and unbiased state change patterns along a B cell lineage. Each simulation resulted in a dataset of BCR sequences, each associated with a single trait value (A or B) resulting from the simulation process. The goal of our simulation analysis is to determine if the summary statistics provide useful information about the mode and tempo of trait evolution.

We ran 20 simulation repetitions for each parameter combination, and tested the significance of each of the proposed statistics to assess their statistical power. Our simulations are designed to generate trees whose tip-states are more clustered together than if the tips states are randomly distributed across the tree tips. Consistent with this expectation, 320/320 simulation repetitions in which r < 1000 (*i.e.* overall rate of state change < 1000 changes/mutation/ site) showed a significantly low *PS* statistic regardless of other parameters ($\delta < 0$; p < 0.05; S3 Fig). This confirms the *PS* test's usability for detecting nonrandom association between tree topology and trait values. However, at r = 1000, only 3/80 repetitions showed a significantly low *PS* statistic (S3 Fig), indicating this relationship is difficult to detect at high rates of state change.

We used the same simulations to test whether the *SC* statistic was capable of detecting the direction of state changes in B cell repertoires. A total of 300 simulation repetitions were performed using parameters expected to give biased (directed) state changes; namely, with lineages always beginning in A ($\pi_a = 1$) and/or highly biased rates of state change from A to B ($r_{ab} = 10$). Surprisingly, only 3/300 of these simulations showed a significantly high *SC* from A to B ($\delta > 0$; p < 0.05; S4 Fig). By contrast, 186/300 showed a significantly low *SC* from A to B ($\delta < 0$; p < 0.05). This indicates that significantly high *SC* statistics are highly conservative, while significantly low *SC* statistics are primarily driven by overall phylogenetic association with a trait. This issue is likely exacerbated as dataset size grows, hence the *SC* test is likely still useful for single lineages [24,25] or for detecting very strong trends in large datasets [6]. However, given these results the *SC* test does not appear appropriate as a general solution for detecting biased migration and differentiation in B cell repertoire datasets.

We next tested whether biased state change patterns were detected by the *SP* test. To test this method's false positive error rate, we first investigated simulations with totally unbiased state changes; namely, in which lineage trees were equally likely to begin at state *A* or *B* ($\pi_a = 0.5$) and had equal rates of state changes between *A* and *B* ($r_{ab} = r_{ba} = 1$). *SP* tests from *A* to *B* on these datasets resulted in a roughly uniform distribution of *p* values across all tested migration rates ($\delta > 0$, p < 0.05 in 5/100; Fig 2A). This indicates that completely unbiased state changing is consistent with the null hypothesis of this test. Simulations in which lineages always had state *A* at the root ($\pi_a = 1$) and/or the relative rate of state change was higher from *A* to *B* ($r_{ab} = 10$) were expected to give high *SP* statistics. At low overall rates of state change (r = 10), 55/60 of these simulations had significantly high *SP* statistics from *A* to *B* ($\delta > 0$; p < 0.05; Fig 2B–2D). As the rate of state change increased (r = 25, 50, 100, or 1000), this relationship diminished as the distribution of trait values became less distinguishable from random association (Fig 2B–2D). These results indicate that, under this two state Markov model framework, a significantly high *SP* statistic is associated with biased origination, biased rate of state change, or both depending on the overall rate.

Finally, we used these simulated datasets to test whether the *SP* test is affected by biased data sampling, as this potential bias is important for some other phylogeographic methods of trait evolution e.g. [37]. We tested this by randomly discarding half of the sequences associated with *A* in simulations with totally unbiased state change ($\pi_a = 0.5$, $r_{ab} = 1$). Though *SP* tests from *A* to *B* on these datasets gave a uniform *p* value distribution when all sequences were included (Fig 2A), *SP* statistics from *A* to *B* became significantly high when half of *A* sequences were discarded (Fig 2E). This indicates that severely biased sampling may give a similar signature as biased origination or state change for the *SP* test (Fig 2B–2D). Biased sampling may be caused by a variety of experimental factors, and applications of these statistics to empirical datasets will need to carefully consider possible effects of biased data collection for each trait type.

Differentiating complex relationships among trait values

All the tests detailed above are extendable to data with more than two states; however, due to its superior performance in two state simulations, we will focus in the rest of this study on the



Fig 2. Distribution of *SP* test *p* values from *A* to *B* from two state simulation analyses in which state change between state *A* and *B* was determined by the probability of starting in *A* (π_a), relative rate of migrating from *A* to *B* (r_{ab}), and the average rate of state change (*r*). To the left of each plot, possible starting states are circled, relative rates are shown by arrowhead size. (a) $\pi_a = 0.5$, $r_{ab} = 1$, fully unbiased state change, shows roughly uniform distribution of *p* values at all tested rates. (b) $\pi_a = 0.5$, $r_{ab} = 10$ shows low *p* values at rates < 50. (c) $\pi_a = 1$, $r_{ab} = 1$ shows low *p* values at low rates (10) but not at higher rates. (d) $\pi_a = 1$, and $r_{ab} = 10$ shows low *p* values at rates < 100. (e) $\pi_A =$ 0.5, $r_{ab} = 1$ shows low *p* values at rate < 50 if 50% of *A* sequences are discarded. Compared to (a), this shows that *p* values are sensitive to biased sampling of sequences. Red lines show the cutoff of *p* value = 0.05.

https://doi.org/10.1371/journal.pcbi.1009885.g002

SP test. The permutation step of the SP test usually permutes trait values within each tree separately (Methods). However, when more than two states are present it may be advantageous to randomize trait value assignments among trees rather than just within each tree. This changes the null hypothesis, which is now that the proportion of state changes observed is the same as that expected if trait values are randomly distributed among all trees. Deviations from this null hypothesis may be due not only to biased ancestor/descendant relationships within individual trees, but also co-occurrence of trait values within different trees. To demonstrate the difference between these two mechanisms, we performed simulations with four trait values: A, B, C, and D. To test the difference between simple association and biased ancestry, these simulations used unbiased state change between A and B, and unidirectional state change from C to D. Trees began with states A, B, or C in equal probability; state changes were allowed in both directions from A to B and unidirectionally from C to D. For each repetition, the rate of state change r = 10, and relative rates were equal among allowed state changes. Performing the SP test on these simulations while permuting among trees showed significantly higher SP statistics in both directions between A and B, and between C and D than expected (20/20 for each; Fig 3A). This indicates the SP test when permuting among trees detected the association between these trait values but not the directionality of C to D state changes. In contrast, the SP test when permuting only within trees correctly yielded a significantly high SP statistic from C to D in 20/20 simulations; further, no simulation yielded a significantly high SP statistic from D to C, indicating a low false positive rate (Fig 3B). No simulation using either permutation method showed a significantly high SP statistic between unassociated trait values (e.g. A and C), indicating a low false positive rate. These results indicate that permuting trait values within trees is



Fig 3. Distribution of *SP* test *p* values from four state simulation analyses under multiple modes of evolution diagrammed to the left of each plot. Twenty repetitions were performed in each scenario. In simulations, possible starting states are circled and possible state changes are shown with arrows. All allowed state changes occurred at the same relative rate and the total rate of state change (*r*) was 10 changes/mutation/site (see Fig 2). (a) Permuting trait values among trees reveals low *p* values for all state changes between *A* and *B*, and between *C* and *D*. (b) Permuting within each tree reveals low *p* values from *C* to *D*, but not between *A* and *B*. Both **a** and **b** imposed no constraints on the types of state changes allowed in the maximum parsimony algorithm. (c) Direct switching simulations result in low *p* values from *A* to *D*, but not from other states to *D*. (d) Sequential switching simulations result in low *p* values from *B* and *C* to *D*, but not from *A*. (f) Unconstrained switching simulations also result in low *p* values from *B* and *C* to *D*, but not from *A*. The strange results of **e** and **f** are likely artefacts of the constrained parsimony algorithm, which forbids reverse alphabetical state changes (e.g. *D* to *C*), used to count state changes in simulations **c-d**.

https://doi.org/10.1371/journal.pcbi.1009885.g003

a more effective means of detecting biased ancestor/descendant relationships, while permuting between trees is more appropriate for detecting associations among traits.

Differentiating constrained modes of state change

In some instances, there are known constraints to the direction that state changes can occur, such as in isotype switching. Isotype-determining constant regions in humans are ordered as IgM/IgD, IgG3, IgG1, IgA1, IgG2, IgG4, IgE, IgA2. Human B cells begin with IgM/IgD, and because the mechanism of class switching is irreversible, these events can only occur sequentially in the order specified. For instance, IgA1 can switch to IgG4, but not to IgM or IgG1. This constraint may be naturally incorporated into the Sankoff parsimony algorithm [29] by making impossible isotype switches have an arbitrarily high weight. A frequent focus of isotype

switching analysis is whether a particular isotype (e.g. IgE) arises from direct switching from IgM or from sequential switching from an intermediate isotype [9,10]. These types of hypotheses could be investigated using the *SP* test.

To determine if the SP test can usefully distinguish between types of constrained relationships among trait values, we simulated datasets to represent possible isotype switching patterns. As above, datasets contained four trait values: A, B, C, and D under different modes of evolution. Because questions often focus on the origin of a particular isotype [10] we only counted state changes leading to D when calculating SP statistics. Further, because state changes can only occur in a particular direction, we permute trait values among trees in these tests to increase power. While we previously showed that permuting among trees confuses biased association with biased ancestry (Fig 3A), switching between these states can only occur in one direction. Because of this, association between two states implies a direction of switching and among tree permutation is justifiable. We first simulate direct switching in which trees always had state A at the root and only state changes from A to the other states were allowed (Fig 3C). We expected these simulations to show a significantly high SP statistic only from A to D. Confirming this expectation, all 20 of these simulations had a high SP statistic from *A* to *D* (δ > 0, *p* < 0.05; Fig 3C). We next simulated sequential switching, where arriving at state D requires transitioning through B. All trees began in A and state changes were allowed from A to B and C, but D arose only from B. We expected these simulations to show a significantly high SP statistic only from B to D. All 20 of these simulations showed a significantly high *SP* statistic from *B* to D ($\delta > 0$, p < 0.05; Fig 3D). These results demonstrate that the SP test using constrained parsimony can discriminate between simple hypotheses of isotype switch patterns, such as direct versus sequential switching.

We next investigated whether the *SP* test can distinguish between more complex types of constrained switching. We simulated irreversible isotype switching in which trees begin with state *A*, and only state changes moving alphabetically (*A* to *D*) were allowed. Naively, we may expect these simulations should show similar *SP* test results from *A*, *B*, and *C* to *D*. However, all 20 of these simulation repetitions showed a significantly high *SP* statistic to *D* from *B* and *C*, but not from *A* (**Fig 3E**). As a control, we simulated unconstrained switching in which trees begin randomly at any state and may change between all states. Using a constrained parsimony model, these simulations showed the same significantly high *SP* to *D* from *B* and *C*, but not from *A* (**Fig 3F**), indicating that this pattern is possibly an artifact of the constrained parsimony model. These results demonstrate that, while the *SP* test outlined here can distinguish between simple types of constrained state change, its relationship to more complex modes of constrained state change such as irreversible evolution are difficult to predict, and should be interpreted cautiously.

Controlling false positives in the case of slow rates of state change

Simulations based on empirical data have shown that the *SP* test can reliably detect biased state change along phylogenetic trees in many cases (**Figs 2** and **3**). However, because the null hypothesis of the *SP* test is that trait values are randomly distributed along tree tips, the *SP* test may have a high false positive rate in situations where the total number of predicted state changes along the tree is significantly different than expected from the null hypothesis. One such case is that of slow state change along a large tree. We confirmed this using 50 repetitions of two-state simulations using unbiased ($\pi_a = 0.5$, $r_{ab} = 1$) switching along a single ladder phylogeny of 1000 tips and a slow rate of state change r = 1 state change/mutation/site (Methods, S6 Fig). Though the switching process in these trees was unbiased, 42% of repetitions had a significantly high *SP* statistic from *A* to *B*, and 32% had a significantly high *SP* statistic from *B*

to *A*, for a total false positive rate of 74%. This indicates that, without correction, the *SP* test is unreliable when the rate of state change is slow and the tree is large.

We investigated whether the issue of slow state change rates over large trees was a significant problem in empirical datasets by characterizing the ratio of tips-to-state transitions for each tree. Slow rates of state transition are expected to produce fewer state transitions along a tree. Simulations using ladder phylogenies confirm this prediction: simulations using a slow rate of state change (r = 1) had a mean tip-to-state change ratio of 326.8; however, when r = 10this ratio dropped to a mean of 37.6, and when r = 100 this ratio was only 5.2 (S7 Fig). We next calculated the tip-to-state change ratio in three empirical datasets in lineages with at least 20 sequences. The first of these characterized cell type transitions in HIV infection, and had a mean tip-to-state change ratio of 33.4. The second empirical dataset focused on isotype switching in a food allergic patient, and had a mean tip-to-state change ratio of 11.1 (S7 Fig). Finally, we included a recently published dataset of B cells obtained from blood and thymus samples from myasthenia gravis patients [8], which had a mean tip-to-state change ratio of 35.4 (S7 Fig). These results show that rates of state change in empirical BCR datasets are characteristic of simulations in which r is between 10 and 100 state changes/mutation/site. The SP test performs well under these rates (Fig 2). Collectively, these results show that trees with rates of state change that are problematically slow for the SP test are identifiable in practice by calculating the ratio of tips to predicted state changes, and uncharacteristic of the BCR datasets analyzed in this study.

To further address the issue of poor performance during slow rates of state change, we developed a simple down-sampling algorithm to control the false positive rate when the rate of state change (r) is low (Methods). This algorithm randomly down-samples each clonal lineage such that its ratio of tips to state changes is less than or equal to a pre-specified threshold. To assess whether this controlled the false positive rate, we first performed simulations in which the state change process was completely unbiased. In these simulations, the proportion of repetitions with a significantly high SP test statistic from A to B indicated the rate of false positives from A to B. As expected, the false positive rate was high when the ratio of tips to state changes was high. At a tip-to-state change ratio of 500, 32% of repetitions simulated with r = 1 had a significantly high SP test statistic from A to B in repertoires with 100 trees (Fig 4A). When these same lineages were down-sampled to a maximum ratio of 10, however, the false positive rate from A to B was only 2%, less than expected given the alpha value of 0.05. To test if downsampling to a low tip-to-state change ratio led to a significant reduction in power, we repeated these analyses with simulations in which the relative rate of switching from A to $B(r_{ab})$ was 10. In these simulations, repetitions with significantly high SP statistics from A to B represent true positives. We found that at all levels of down-sampling, simulations with biased ($r_{ab} = 10$) and slow rates of state change (r = 1 and 10) had true positive rates ranging from 82% to 100% in repetitions containing at least 20 lineages (Fig 4B). This demonstrated that down-sampling did not significantly lower power in these cases. Down-sampling to a ratio of 10-20 did have low power when performing analyses in faster rates of state change r = 100 (mean true positive = 6.0%) or when there was only one tree within a simulated repertoire (mean true positive = 14.7%). Down-sampling also did not increase the false positive rate of SP test from B to A in the simulations tested (S8 Fig). Repeating previous simulations (Fig 2) with down-sampling to a tip-to-state change ratio of 20 showed consistent results, with a reduction in power at higher rates of state change (S9 Fig). Collectively, these results suggest that ensuring all trees analyzed have a maximum tip-to-state change ratio of 10-20 adequately controls the false positive rate in the case of slowly changing discrete states without severely limiting power. This could be achieved by down-sampling lineages with a higher ratio of tips to state changes.



Fig 4. Controlling false positive rates using down-sampling. Y axis shows the proportion of 50 simulation replicates in which the *SP* test from *A* to *B* was significant. Simulations were performed on large ladder phylogenies (Methods) at varying rates, biases, and trees per repetition (representing lineages within a simulated repertoire). In each case, simulated lineages were down-sampled to the specified maximum tip-to-state change ratio. (a) Unbiased simulations, significant *SP* test values indicate false positives. These show that the *SP* test has a high rate of false positives when the rate of state change is low (i.e. 1 change/mutation/site) and the maximum tip-to-state change ratio is high (> 20). Down-sampling lineages to a maximum tip-to-state change ratio of 10–20, however, controls this false positive rate. (b) Simulations with rates biased from *A* to *B*, in which significant *SP* test values indicate true positives. These results show that down-sampling does not simply reduce power. However, power is lowered when the switching rate is high (>100) or the number of trees/repertoire is low (< 20).

https://doi.org/10.1371/journal.pcbi.1009885.g004

Differentiation of B cell subtypes during HIV infection

Over the course of the immune response, B cells differentiate into multiple cellular subsets with distinct properties. Recent studies have focused on the role of T-bet, a transcription factor usually associated with differentiation of T cells, in shaping B cell responses during infection. For example, [6] used data from three HIV+ patients to demonstrate that $CD19^{hi}$ memory B cells ($CD19^{hi}$ MBCs, a surrogate for T-bet+ B cells) represented earlier states in the affinity maturation process than germinal center B cells (GCBCs), and to define the relationships among other B cell subtypes including $CD19^{lo}$ MBCs and un-switched MBCs. More specifically, [6] used the *SC* test with trait values permuted among trees. However, the simulation analyses performed here demonstrated the *SC* test is highly conservative, and that permuting among trees may only detect unstructured association among trait values (**Fig 3**). It is therefore not clear whether the relationship from $CD19^{hi}$ MBCs to GCBCs observed in [6] was driven by biased ancestor/descendant relationships among these cell types within trees. Our results

above suggests that the *SP* test using within tree permutation would be a more appropriate test of this relationship.

We characterized the relationships among B cell subsets with a two-tailed *SP* test using within tree permutation for each of the three subjects. These analyses showed a significantly high *SP* statistic from CD19^{hi} MBCs to GCBCs in all three subjects, and to CD19^{lo} MBCs in one subject ($\delta > 0$, p < 0.025; **Figs 5B–5D and S10**). These analyses confirm the conclusions in [6] that CD19^{hi} MBCs are significantly closer, cladistically, to the predicted germline sequence than GCBC sequences than expected by chance. Naively, one may interpret this as evidence that GCBCs derive from CD19^{hi} MBCs. However, because GCBCs are expected to have far higher mutation rates than MBCs, the observed patterns are also consistent with early production of CD19^{hi} MBC from GCBCs, followed by a near cessation of mutations in CD19^{hi} MBCs. This is consistent with the conclusions of [6] that CD19^{hi} MBCs represent earlier stages in the GC reaction, rather than the direction of differentiation. Overall, these *SP* tests confirm that the previously observed relationships from CD19^{hi} MBCs and CD19^{lo} MBCs to GCBCs are driven by biased ancestor/descendant relationships within trees rather than simply association in the same trees, as may have been the case from the previously used *SC* tests with among tree permutations [6].



Fig 5. Analysis of B cell subtypes in three HIV+ subjects. (a) Example tree visualized using *ggtree* [38,39] showing observed relationship between CD19^{hi} MBCs and GCBCs. (**b-d**) Direction of significant *SP* test δ values for subjects 1 (**b**), 2 (**c**), and 3 (**d**). Arrows within each diagram show the direction of significantly high (blue) or significantly low (red) *SP* statistics between CD19^{hi} MBCs, CD19^{lo} MBCs, unswitched MBCs (Unsw), and GCBCs in each subject.

https://doi.org/10.1371/journal.pcbi.1009885.g005

Sequential isotype switching to IgE and IgG4

Antibody isotypes are a major determinant of function. Of particular interest is characterizing whether IgE antibodies, the primary antibody isotype associated with allergic response, arise directly from IgM switching, or through sequential switching from another downstream isotype [9,10]. Previous studies have shown evidence that IgE in mice and human adults arises from sequential switching primarily from IgG [9,10], though a recent study in 27 humans in the first three years of life found evidence of a greater association between IgA1 and IgE in children with food allergy and eczema [36]. Specifically, [36] showed a higher number of shared clones between IgE and IgA1 than between IgE and other isotypes in these subjects. A phylogenetic test of this relationship would confirm that IgE and IgA1 sequences show a direct ancestor-descendant relationship within these B cell trees rather than just being part of the same clone.

We applied our discrete trait framework to determine the origins of IgE in a single subject (id = 2442) from [36]. This subject was selected due to reported history eczema, food allergy, and B cell clones containing IgE and other isotypes [36]. Using an *SP* test in which only state changes leading to IgE were considered and trait values were permuted among trees, we found a significantly high *SP* statistic from IgA1 to IgE (Fig 6B). No other isotype showed a



Fig 6. Analysis of antibody isotypes from a single subject. (a) Example trees visualized using *ggtree* [38,39] showing observed relationships between cells expressing BCRs with IgA1 and IgE, as well as IgG1 and IgG4 isotypes. IgE and IgG4 are indicated on each tree using larger tip circles. (b) Distribution of *SP* test δ values to IgE from each of the other isotypes (different colors), and distribution of *SP* test δ values to IgG4 from each of the other isotypes (different colors). State changes in **b** were calculated using constrained parsimony which forbids state changes that violate the geometry of the Ig heavy chain locus, and *SP* tests were performed using permutation among trees.

https://doi.org/10.1371/journal.pcbi.1009885.g006

significantly high *SP* statistic to IgE. These results favor IgE arising from sequential switching through IgA1 over direct switching from IgM in this subject. Performing a similar test using only state changes leading to IgG4 revealed a significantly high *SP* statistic from IgG1, IgG2, and IgG3 to IgG4 (**Fig 6B**). This pattern is similar to irreversible switching within the IgG family (**Fig 3E**). As shown in simulation analyses, this test is not suited to infer relative rates of switching from different isotypes if all kinds of switches are considered. However, these results are most consistent with origin of IgG4 through sequential switching from IgA1. Overall, these results are consistent with the conclusions of [36] that IgE arose preferentially through sequential switching from IgA1 in this subject. Our results further suggest IgG4 arose preferentially through sequential switching from IgG subtypes in this subject.

Discussion

Phylogenetic techniques have the potential to reveal important information about B cell migration, cellular differentiation, and isotype switching. While great strides have been made using phylogenetic models to study evolution and trait change generally, there are significant challenges to translating these approaches to B cells. As a step in this direction, hypotheses about the ancestor/descendant relationships of B cell trait values may be usefully investigated using heuristic approaches that are robust to uncertainties in branch length estimation. Here, we outline three maximum parsimony-based summary statistics (two of which were used in prior studies) to characterize the distribution of trait values along phylogenetic trees. Significance of all of these statistics is calculated using a permutation test. We demonstrate the efficacy of these tests using simulations, and show that the *SP* test is the most useful for characterizing ancestor/descendant relationships among trait values. We further demonstrate how these statistics can test hypotheses about empirical B cell datasets by characterizing the relationship between T-Bet+ memory B cells and germinal center B cells in three HIV+ patients, and the class switching origins of IgE and IgG4 in a human subject during the first three years of life.

Simulations demonstrate that the *SP* test was uniquely able to determine the direction of biased origination and state change among the approaches investigated. In simple simulations containing two states (*A* and *B*) a significantly high *SP* statistic from *A* to *B* was associated with origination in *A* and biased state change from *A*. This signal decreased as the overall rate of switching increased. In more complex scenarios, the *SP* test was able to differentiate between traits that were generated through biased state change in a particular direction versus traits that were simply associated with each other. The *SP* test was also able to distinguish between simple modes of constrained evolution such as direct and sequential switching. These results indicate the *SP* test may have broad utility in characterizing ancestor/descendant relationships among B cell discrete traits.

We next used two datasets to demonstrate that the *SP* test could be used to derive meaningful biological conclusions. In the first, we confirm that T-Bet+ memory B cells tend to be the predicted immediate ancestors of GC B cells within lineages trees obtained from three HIV+ subjects. Though this relationship may primarily be due to differences in mutation rate over time between memory and GC B cell subsets, this does confirm prior findings and demonstrates that the T-Bet+ memory B cell subset represents an earlier state in the affinity maturation process, possibly contributing to an impaired immune response to HIV [6]. We next characterized the isotype switching patterns of sequences obtained from a human over the first three years of life [36]. In this analysis, we found evidence of sequential switching from IgA1 to IgE, as well as evidence of sequential switching from IgG subtypes to IgG4. Sequential switching from IgA1 to IgE is consistent with [36] but not other analyses performed on data taken from adults, which favor sequential switching from IgG [10]. This possibly reflects differences in isotype switching patterns between adults and children. Overall these results demonstrate that the discrete trait analysis framework developed here can be used to test important hypotheses about B cell differentiation and class switching.

There are a number of limitations with these methods. First, tree topologies were estimated using maximum parsimony. While maximum parsimony is not a statistically consistent estimator of tree topology and is known to give inaccurate predictions over long branch lengths [40], it has been shown to be an accurate estimator of tree topology in certain B cell applications [27], and is widely used in B cell phylogenetic analysis [5,22]. In any case, the statistics presented here are not limited to tree topologies inferred through maximum parsimony, and parsimony was used primarily for computational expediency. The three statistics proposed here (Eqs 1-3) are also based on maximum parsimony, and may have similar inaccuracies over long branch lengths. Further, the statistical tests assume that the process of state change is independent of the tree shape, when the two may be coupled e.g. [41]. This assumption of independence is commonly made in discrete trait analysis e.g. [13] to enable computational tractability, and because the actual link between tree shape and state change is unknown or cannot be modeled. A significantly high SP statistic could be potentially caused by factors other than biased state change. For instance, because tree branch lengths represent genetic distance rather than time, it is possible that cell types with low mutation rates over time will spuriously appear ancestral to those with high mutation rates. As discussed previously, this effect likely underlies our analysis of B cell subtypes in HIV infection. Finally, it is possible that SHM is actually occurring at another, un-sampled site which is seeding the sites that were sampled (S1 Fig). Overall, it is important to carefully consider alternative explanations when trying to determine the biological basis for a significantly high SP statistic.

An important limitation of the *SP* test is that, like many other phylogeographic approaches e.g. [37], it is affected by biased data sampling. This may arise due to experimental factors that are difficult to control. For instance, under-sampling a trait value may cause a spurious, significantly high *SP* statistic from that trait value. Previous analyses of viral migration have dealt with potential sampling bias by performing tests across multiple down-sampling repetitions [14]. In practice, it can be difficult to know if B cells with certain trait values have been sampled proportionally to their relative population sizes. However, if a type of B cell is known to be under-sampled in a particular experiment, and is predicted to be the descendant of another B cell type, it can be argued that this relationship is unlikely to be due to biased sampling (Fig 2E). Alternatively, if multiple samples are tested it is possible that these samples will have a wide range of sequence proportions belonging to different traits. If these differences in sequence proportions are uncorrelated with *SP* test results, it could be argued that observed results are unlikely to be due to consistent under-sampling of B cells with a particular trait value.

Our simulation analyses revealed that the *SP* test is difficult to interpret when considering complex constrained models such as irreversible isotype switching (**Fig 3**). To recreate isotype switching, we performed four state simulations in which only state changes proceeding in the direction of state *A*, *B*, *C*, and *D* were allowed. Unexpectedly, these simulations tended to show a significantly low *SP* statistic from *A* to *D*, but a significantly high *SP* statistic for *B* and *C* to *D* (**Fig 3E**). This biased trend is likely driven by the fact, due to constraints in the direction of state change, randomized trees tend to have more switches from *A* than expected based on the relative frequency of *A*. This produces a significantly high *SP* for switches from *A* to *D*. An alternative may be to use the *SP* statistic (**Eq 3**) without comparing to a null distribution, which is equivalent to comparing the relative frequency of each type of switch observed. However, the observed switch frequency (*SP* statistic) is not proportional to the true relative rate of

state change in general. For instance, in the two state Markov model simulations presented here (**Fig 2**), the *SP* statistic alone is both positively and negatively related to the true relative rate of state change, depending on other parameters (**S5 Fig**). Comparing *SP* statistics to those obtained from randomized trees (i.e. the *SP* test) usefully corrects this relationship in unconstrained models (**Fig 2**), but not always in constrained ones (**Fig 3C-3E**). Ultimately, isotype switching is a complex, constrained process, and our analyses suggest the relative rates of isotype switching inferred from B cell trees should be interpreted cautiously. We suspect a general method for accurately estimating these rates will require a model-based approach, such as a non-reversible Markov model.

Finally, it is important to note that the *SP* test can have a high false positive rate when the rate of state change is very slow relative to the rate of mutations. This is because the null hypothesis of the test is that trait values are randomly distributed along each tree's tips. This is convenient to calculate, but makes the test sensitive to cases where state changes are produced in an unbiased manner but where the total number of state changes along the tree is significantly different than that expected from the null hypothesis. A slow rate of state change along a large tree has this effect, and the *SP* test performs poorly in this case (**Figs 4 and S6**). However, we show that i) trees with slow rates of state change are identifiable in practice by calculating the ratio of tips to predicted state changes (**S6 and S7 Figs**) ii) trees with problematically slow rates of state change are uncharacteristic of the real B cell lineage trees surveyed (**S7 Fig**), and iii) the false positive rate from this source can be controlled using a simple down-sampling algorithm (**Figs 4 and S8 and S9**). In practice, we recommend always down-sampling all trees to a tip-to-state change ratio ≤ 20 . This feature is included in *dowser* v0.1.0.

Future methods to differentiate migration, differentiation, and isotype switching patterns in B cells might improve upon the approach developed here by explicitly modeling these processes along a phylogeny, incorporating branch length information, and better accounting for uncertainty in tree topology. The heuristic approach introduced here crucially does not use branch lengths to help predict internal node states of the tree. Ignoring this source of information likely lowers power, but is possibly advantageous because the relationship between mutation rate and time is not currently well understood, and likely varies by cell type. While the approach developed here uses phylogenetic bootstrap replicates to account for uncertainty in tree topology [31], this may also be done using a posterior distribution of topologies generated by MCMC sampling. This was recently done for naive sequence inference in individual B cell lineages [42]. Phylogenetic bootstrapping has less desirable statistical properties than posterior distributions, but is a widely used means of assessing reproducibility of tree topology and is more computationally tractable for large datasets. Overall, though there is potential for improvement, the approach introduced here effectively deals with important challenges such as incorporating information across trees, accounting for uncertainty in tree topology, and scaling efficiently when analyzing large datasets.

A phylogenetic discrete trait analysis framework fills an important gap in B cell sequence analysis. The proposed framework provides a principled, flexible, and scalable approach for characterizing migration, cellular differentiation, and isotype switching in a wide array of contexts. This differs from other phylogenetic tools we developed recently, which used modelbased approaches for characterizing somatic hypermutation and clonal selection [4,28].

Supporting information

S1 Appendix. Maximum ambiguity resolution of polytomies. (PDF)

S2 Appendix. Processing of empirical datasets. (PDF)

S1 Fig. Possible mechanisms of tracking B cell migration through somatic hypermutation. (a) Direct migration from tissue A, followed by extrafollicular mutation in tissue B. (b) Migration from tissue A, through a germinal center where SHM accumulates, and then migration to tissue B. (c) Early germinal center exit to tissue A, late germinal center exit to tissue B. This is not actually migration between tissue A and B. (d) Inferred phylogenetic tree in all three cases showing tissue A as the likely ancestral state compared to tissue B. (PDF)

S2 Fig. Visual diagram of *SP* test calculation on a single lineage tree. (PDF)

S3 Fig. Simulation analyses with *PS* **test.** Distribution of *PS* test *p* values for the hypothesis that $\delta < 0$ from two state simulation analyses. See Fig 2 for analysis of the same data with the *SP* test. In these simulations, change between state *A* and *B* was determined by the probability of starting in *A* (π_a), relative rate of migrating from *A* to *B* (r_{ab}), and the average rate of state change (*r*). To the left of each plot, possible starting states are circled, relative rates are shown by arrowhead size. (a) $\pi_a = 0.5$, $r_{ab} = 1$, fully unbiased state change. (b) $\pi_a = 0.5$, $r_{ab} = 10$. (c) $\pi_a = 1$, $r_{ab} = 1$. (d) $\pi_a = 1$, and $r_{ab} = 10$. (e) $\pi_A = 0.5$, $r_{ab} = 1$, 50% of *A* sequences are discarded. Red lines show the cutoff of *p* value = 0.05. (PDF)

S4 Fig. Simulation analyses with *SC* **test.** Distribution of *SC* test *p* values for the hypothesis that $\delta > 0$ from two state simulation analyses. See Fig 2 for analysis of the same data with the *SP* test. In these simulations, change between state *A* and *B* was determined by the probability of starting in *A* (π_a), relative rate of migrating from *A* to *B* (r_{ab}), and the average rate of state change (*r*). To the left of each plot, possible starting states are circled, relative rates are shown by arrowhead size. (a) $\pi_a = 0.5$, $r_{ab} = 1$, fully unbiased state change. (b) $\pi_a = 0.5$, $r_{ab} = 10$. (c) $\pi_a = 1$, $r_{ab} = 1$. (d) $\pi_a = 1$, and $r_{ab} = 10$. (e) $\pi_A = 0.5$, $r_{ab} = 1$, 50% of *A* sequences are discarded. Red lines show the cutoff of *p* value = 0.05. (PDF)

S5 Fig. *SP* statistics in two state simulation analyses. Distribution of raw *SP* statistics from *A* to *B* in two state simulations. The red line is at 0.5, showing equal switch frequency. See Fig 2 for analysis of the same data with the full *SP* test. In these simulations, change between state *A* and *B* was determined by the probability of starting in *A* (π_a), relative rate of migrating from *A* to *B* (r_{ab}), and the average rate of state change (r). To the left of each plot, possible starting states are circled, relative rates are shown by arrowhead size. (a) $\pi_a = 0.5$, $r_{ab} = 1$, fully unbiased state change. (b) $\pi_a = 0.5$, $r_{ab} = 10$. (c) $\pi_a = 1$, $r_{ab} = 1$. (d) $\pi_a = 1$, $r_{ab} = 10$. (e) $\pi_A = 0.5$, $r_{ab} = 1$, 50% of *A* sequences are discarded. Note that at low rates (r = 10), origination at *A* (c and d) shows *SP* statistic > 0.5. However, at higher rates (r > 10), increased rate of state change from *A* to *B* actually produces lower *SP* statistics. This effect increases as overall rate (r) increases. This indicates that the *SP* statistic itself is not a good estimator for the relative rate of state change, and is why we only interpret p values resulting from comparison of the *SP* statistic to a null distribution (i.e. the *SP* test).

(PDF)

S6 Fig. Simulations across large ladder phylogenies with slow rates of state change. Five randomly chosen phylogenies from simulations using ladder phylogenies of 1000 tips, unbiased rates of state change ($r_{ab} = 1$) and r = 1 state change/mutation/site. These trees appear as

diagonal lines due to their size. For ease of visualization, tips at state A (middle) and B (left) are shown separately. Single dots usually represent changes along tip branches. For each tree, the observed and permutated SP test statistics from A to B (blue line/green histogram, respectively) and B to A (red line/orange histogram, respectively) are shown on the right. These confirm that the uncorrected SP test has a high rate of false positives across large trees with slow rates of state change.



S7 Fig. Tip-to-state change ratio of simulations and empirical BCR datasets. The tip to state change ratio was calculated for each lineage with at least 20 unique sequences in each dataset. For computational efficiency, all lineages were sampled to a maximum tip-to-state change ratio of 500. The top three panels show results from the first 5 repetitions of simulations with large ladder phylogenies used in Figs 4 and S6. These simulations used unbiased state change ($\pi_a = 0.5$, $r_{ab} = 1$) and contained 100 trees per repetition. Slow (r = 1) rates of state change give high tip to state change ratios, while faster simulations (r = 10 or 100) have much lower ratios. The bottom three panels show results from three empirical datasets. The fourth panel from the top is the HIV dataset (Fig 5) containing only CD19^{hi} and GCBC cells [6]. The fifth is the isotype dataset (Fig 6) [36]. The bottom panel is previously processed data from a recent study containing thymus and blood samples from myasthenia gravis patients [8]. In all three cases, the tip-to-state change ratio of empirical BCR data is more characteristic of simulations in which r = 10 changes/mutation/site. The SP test performs well under these rates (Fig 2,4 and S9).



S8 Fig. Down-sampling simulation analysis. Simulations are the same as in **Fig 4**, but the proportion of significant *SP* tests from *B* to *A* rather than *A* to *B* are shown. (a) Unbiased simulations, significant *SP* test values indicate false positives, as in **Fig 4**. (b) Simulations with rates biased from *A* to *B*. As expected, there is no appreciable rate of significant *SP* statistics from *B* to *A* in this scenario.

(PDF)

S9 Fig. Repeating previous two-state simulations with down-sampling. *SP* test results performed on the same simulation data as in **Fig 2**, but with each lineage down-sampled to a maximum tip to state change ratio of 20. Results are largely unchanged except for slight reduction in power at higher rates of state change. (PDF)

S10 Fig. SP test δ values for differentiation of B cell subtypes within HIV+ subjects. Distribution of SP test δ values between all four B cell subtypes included (different colors) for each subject. (PDF)

Author Contributions

Conceptualization: Kenneth B. Hoehn, Steven H. Kleinstein. Data curation: Kenneth B. Hoehn. Formal analysis: Kenneth B. Hoehn. Funding acquisition: Steven H. Kleinstein. Investigation: Kenneth B. Hoehn. Methodology: Kenneth B. Hoehn, Oliver G. Pybus, Steven H. Kleinstein.

Resources: Steven H. Kleinstein.

Software: Kenneth B. Hoehn.

Supervision: Steven H. Kleinstein.

Writing - original draft: Kenneth B. Hoehn, Steven H. Kleinstein.

Writing - review & editing: Kenneth B. Hoehn, Oliver G. Pybus, Steven H. Kleinstein.

References

- 1. Murphy K, Travers P, Walport M, Janeway C. Janeway's immunobiology. New York: Garland Science; 2012.
- Shlomchik MJ, Marshak-Rothstein A, Wolfowicz CB, Rothstein TL, Weigert MG. The role of clonal selection and somatic mutation in autoimmunity. Nature. 1987; 328: 805–811. <u>https://doi.org/10.1038/ 328805a0 PMID: 3498121</u>
- Liao H-X, Lynch R, Zhou T, Gao F, Alam SM, Boyd SD, et al. Co-evolution of a broadly neutralizing HIV-1 antibody and founder virus. Nature. 2013; 496: 469–476. <u>https://doi.org/10.1038/nature12053</u> PMID: 23552890
- Hoehn KB, Heiden JAV, Zhou JQ, Lunter G, Pybus OG, Kleinstein SH. Repertoire-wide phylogenetic models of B cell molecular evolution reveal evolutionary signatures of aging and vaccination. PNAS. 2019; 201906020. https://doi.org/10.1073/pnas.1906020116 PMID: 31636219
- Stern JNH, Yaari G, Heiden JAV, Church G, Donahue WF, Hintzen RQ, et al. B cells populating the multiple sclerosis brain mature in the draining cervical lymph nodes. Sci Transl Med. 2014; 6: 248ra107-248ra107. https://doi.org/10.1126/scitranslmed.3008879 PMID: 25100741
- Austin JW, Buckner CM, Kardava L, Wang W, Zhang X, Melson VA, et al. Overexpression of T-bet in HIV infection is associated with accumulation of B cells outside germinal centers and poor affinity maturation. Science Translational Medicine. 2019;11. <u>https://doi.org/10.1126/scitranslmed.aax0904</u> PMID: 31776286
- Horns F, Vollmers C, Croote D, Mackey SF, Swan GE, Dekker CL, et al. Lineage tracing of human B cells reveals the in vivo landscape of human antibody class switching. eLife. 2016; 5: e16578. <u>https:// doi.org/10.7554/eLife.16578 PMID: 27481325</u>
- Jiang R, Hoehn KB, Lee CS, Pham MC, Homer RJ, Detterbeck FC, et al. Thymus-derived B cell clones persist in the circulation after thymectomy in myasthenia gravis. Proceedings of the National Academy of Sciences. 2020;117: 30649–30660. https://doi.org/10.1073/pnas.2007206117
- He J-S, Subramaniam S, Narang V, Srinivasan K, Saunders SP, Carbajo D, et al. IgG1 memory B cells keep the memory of IgE responses. Nature Communications. 2017; 8: 641. <u>https://doi.org/10.1038/</u> s41467-017-00723-0 PMID: 28935935
- Looney TJ, Lee J-Y, Roskin KM, Hoh RA, King J, Glanville J, et al. Human B-cell isotype switching origins of IgE. Journal of Allergy and Clinical Immunology. 2016; 137: 579–586.e7. https://doi.org/10. 1016/j.jaci.2015.07.014 PMID: 26309181
- Hoh RA, Joshi SA, Lee J-Y, Martin BA, Varma S, Kwok S, et al. Origins and clonal convergence of gastrointestinal IgE+ B cells in human peanut allergy. Science Immunology. 2020;5. <u>https://doi.org/10.1126/sciimmunol.aay4209 PMID: 32139586</u>
- Elsner RA, Shlomchik MJ. Germinal Center and Extrafollicular B Cell Responses in Vaccination, Immunity, and Autoimmunity. Immunity. 2020; 53: 1136–1150. https://doi.org/10.1016/j.immuni.2020.11.006 PMID: 33326765
- Lemey P, Rambaut A, Drummond AJ, Suchard MA. Bayesian Phylogeography Finds Its Roots. PLOS Computational Biology. 2009; 5: e1000520. https://doi.org/10.1371/journal.pcbi.1000520 PMID: 19779555
- Faria NR, Rambaut A, Suchard MA, Baele G, Bedford T, Ward MJ, et al. The early spread and epidemic ignition of HIV-1 in human populations. Science. 2014; 346: 56–61. <u>https://doi.org/10.1126/science. 1256739 PMID: 25278604</u>
- Dudas G, Carvalho LM, Bedford T, Tatem AJ, Baele G, Faria NR, et al. Virus genomes reveal factors that spread and sustained the Ebola epidemic. Nature. 2017; 544: 309–315. https://doi.org/10.1038/ nature22040 PMID: 28405027

- Dellicour S, Baele G, Dudas G, Faria NR, Pybus OG, Suchard MA, et al. Phylodynamic assessment of intervention strategies for the West African Ebola virus outbreak. Nat Commun. 2018; 9: 1–9. https:// doi.org/10.1038/s41467-017-02088-w PMID: 29317637
- Faria NR, Quick J, Claro IM, Thézé J, de Jesus JG, Giovanetti M, et al. Establishment and cryptic transmission of Zika virus in Brazil and the Americas. Nature. 2017; 546: 406–410. <u>https://doi.org/10.1038/</u> nature22401 PMID: 28538727
- Hill SC, Vasconcelos J, Neto Z, Jandondo D, Zé-Zé L, Aguiar RS, et al. Emergence of the Asian lineage of Zika virus in Angola: an outbreak investigation. The Lancet Infectious Diseases. 2019; 19: 1138– 1147. https://doi.org/10.1016/S1473-3099(19)30293-2 PMID: 31559967
- Felsenstein J. Evolutionary trees from DNA sequences: A maximum likelihood approach. J Mol Evol. 1981; 17: 368–376. https://doi.org/10.1007/BF01734359 PMID: 7288891
- Suchard MA, Lemey P, Baele G, Ayres DL, Drummond AJ, Rambaut A. Bayesian phylogenetic and phylodynamic data integration using BEAST 1.10. Virus Evol. 2018; 4. <u>https://doi.org/10.1093/ve/ vey016 PMID: 29942656</u>
- Bouckaert R, Vaughan TG, Barido-Sottani J, Duchêne S, Fourment M, Gavryushkina A, et al. BEAST 2.5: An advanced software platform for Bayesian evolutionary analysis. PLOS Computational Biology. 2019; 15: e1006650. https://doi.org/10.1371/journal.pcbi.1006650 PMID: 30958812
- Ohm-Laursen L, Meng H, Chen J, Zhou JQ, Corrigan CJ, Gould HJ, et al. Local Clonal Diversification and Dissemination of B Lymphocytes in the Human Bronchial Mucosa. Front Immunol. 2018; 9: 1976. https://doi.org/10.3389/fimmu.2018.01976 PMID: 30245687
- Slatkin M, Maddison WP. A cladistic measure of gene flow inferred from the phylogenies of alleles. Genetics. 1989; 123: 603–613. https://doi.org/10.1093/genetics/123.3.603 PMID: 2599370
- Nakano T, Lu L, Liu P, Pybus OG. Viral Gene Sequences Reveal the Variable History of Hepatitis C Virus Infection among Countries. The Journal of Infectious Diseases. 2004; 190: 1098–1108. <u>https:// doi.org/10.1086/422606 PMID: 15319860</u>
- Wallace RG, HoDac H, Lathrop RH, Fitch WM. A statistical phylogeography of influenza A H5N1. PNAS. 2007; 104: 4473–4478. https://doi.org/10.1073/pnas.0700435104 PMID: 17360548
- 26. Felsenstein J. {PHYLIP} (Phylogeny Inference Package) version 3.6a3. 2002.
- Davidsen K, Matsen FA. Benchmarking Tree and Ancestral Sequence Inference for B Cell Receptor Sequences. Front Immunol. 2018; 9. https://doi.org/10.3389/fimmu.2018.02451 PMID: 30429847
- Hoehn KB, Lunter G, Pybus OG. A Phylogenetic Codon Substitution Model for Antibody Lineages. Genetics. 2017; 206: 417–427. https://doi.org/10.1534/genetics.116.196303 PMID: 28315836
- Sankoff D. Minimal Mutation Trees of Sequences. SIAM J Appl Math. 1975; 28: 35–42. <u>https://doi.org/ 10.1137/0128004</u>
- Chen R, Holmes EC. Frequent inter-species transmission and geographic subdivision in avian influenza viruses from wild birds. Virology. 2009; 383: 156–161. https://doi.org/10.1016/j.virol.2008.10.015 PMID: 19000628
- Felsenstein J. Confidence Limits on Phylogenies: An Approach Using the Bootstrap. Evolution. 1985; 39: 783. https://doi.org/10.1111/j.1558-5646.1985.tb00420.x PMID: 28561359
- Parker J, Rambaut A, Pybus OG. Correlating viral phenotypes with phylogeny: Accounting for phylogenetic uncertainty. Infection, Genetics and Evolution. 2008; 8: 239–246. <u>https://doi.org/10.1016/j.</u> meegid.2007.08.001 PMID: 17921073
- Laserson U, Vigneault F, Gadala-Maria D, Yaari G, Uduman M, Heiden JAV, et al. High-resolution antibody dynamics of vaccine-induced immune responses. PNAS. 2014; 111: 4928–4933. https://doi.org/ 10.1073/pnas.1323862111 PMID: 24639495
- 34. Gupta NT, Vander Heiden JA, Uduman M, Gadala-Maria D, Yaari G, Kleinstein SH. Change-O: a toolkit for analyzing large-scale B cell immunoglobulin repertoire sequencing data. Bioinformatics. 2015; 31: 3356–3358. https://doi.org/10.1093/bioinformatics/btv359 PMID: 26069265
- R Core Team. R: A Language and Environment for Statistical Computing. Vienna, Austria: R Foundation for Statistical Computing; 2017. Available: https://www.R-project.org/
- Nielsen SCA, Roskin KM, Jackson KJL, Joshi SA, Nejad P, Lee J-Y, et al. Shaping of infant B cell receptor repertoires by environmental factors and infectious disease. Science Translational Medicine. 2019; 11: eaat2004. https://doi.org/10.1126/scitransImed.aat2004 PMID: 30814336
- Lemey P, Rambaut A, Bedford T, Faria N, Bielejec F, Baele G, et al. Unifying Viral Genetics and Human Transportation Data to Predict the Global Transmission Dynamics of Human Influenza H3N2. PLOS Pathogens. 2014; 10: e1003932. https://doi.org/10.1371/journal.ppat.1003932 PMID: 24586153

- **38.** Yu G, Smith DK, Zhu H, Guan Y, Lam TT-Y. ggtree: an r package for visualization and annotation of phylogenetic trees with their covariates and other associated data. Methods in Ecology and Evolution. 2017; 8: 28–36. https://doi.org/10.1111/2041-210X.12628
- Yu G, Lam TT-Y, Zhu H, Guan Y. Two Methods for Mapping and Visualizing Associated Data on Phylogeny Using Ggtree. Mol Biol Evol. 2018; 35: 3041–3043. https://doi.org/10.1093/molbev/msy194 PMID: 30351396
- Felsenstein J. Cases in which Parsimony or Compatibility Methods Will be Positively Misleading. Systematic Zoology. 1978; 27: 401–410. https://doi.org/10.2307/2412923
- **41.** Hudson RR. Gene genealogies and the coalescent process. Oxford Surveys in Evolutionary Biology. 1991; 7: 1–44.
- Dhar A, Ralph DK, Minin VN, Matsen FA. A Bayesian phylogenetic hidden Markov model for B cell receptor sequence analysis. PLoS Comput Biol. 2020; 16: e1008030. https://doi.org/10.1371/journal. pcbi.1008030 PMID: 32804924